Is financial innovation market-driven or bank-driven? The case of private equity in the 1980s\*

Carolyn Sissoko<sup>†</sup>

17 April 2024

## Abstract:

Financial innovations, such as leveraged buyouts (LBOs), are often categorized as 'market-based' phenomena in order to distinguish them from more traditional 'bank-based' activities. This paper uses call report data to demonstrate the 1980s LBO boom was a bank-driven phenomenon and that the largest US banks relied on the senior loan debt issued in LBOs to provide the earnings that allowed them to absorb the losses from the LDC debt crisis. A banking theory framework explains why bank-driven lending can create economic distortions. One distortion due to this bank finance of buyouts is the increase in corporate income flowing to the financial sector. It remains to be seen whether the corporate debt levels associated with buyouts will turn out to be another distortion. We argue that bank-driven financial innovation is an important framework for studying additional instruments, such as collateralized loan obligations and private credit.

Keywords: leveraged buyout, financial innovation, too big to fail, banking theory

<sup>&</sup>lt;sup>\*</sup> Thanks for very helpful comments go to Dan Awrey, Samule Knafo, Katharina Pistor, Matthias Thiemann, Joscha Wollweber, participants in the Crisis Capitalism Symposium, The Global US Dollar System and the Fed session at EAEPE, and the 6<sup>th</sup> Law and Macro conference. Ayoze Ramirez kindly provided me with an invaluable list of call report categories. Thanks also to the always helpful staff at the Bank of England Archives including Mike Anson and Joseph Hettrick. All errors are of course my own.

<sup>&</sup>lt;sup>†</sup> Senior lecturer, Bristol Business School, University of the West of England. Email: Carolyn.sissoko@uwe.ac.uk

Financial innovations, such as leveraged buyouts or collateralized loan obligations, are often categorized as 'market-based' phenomena in order to distinguish them from more traditional 'bank-based' activities. This essay will study the early history of the buyout industry in detail in order to demonstrate how this distinction can be very misleading. In particular, this essay will set forth the substantial evidence that the leveraged buyout boom of the 1980s was a bank-driven phenomenon, even as it had a profound effect on financial markets, fostering the growth of junk bonds and playing a role in the reduction over time of the importance of the stock market. Understanding the driving role that banks play in the process of financial innovation has important implications for the regulation of innovative financial products today.

The common understanding of the leveraged buyout boom and why it took place was shaped by contemporary finance theory in the 1980s. Jensen and Meckling in 1976 had questioned why corporate shareholders would entrust their savings to corporate managers who were agents that could choose to spend corporate funds in their own interests. They observed that a takeover that bought out all of the shareholders of a publicly traded corporation could serve to align the interests of the owners and the managers. In 1983 Jensen and Ruback studied the evidence on corporate takeovers and concluded that they did in fact result in financial gains to the initial owners of the corporation with no loss to the subsequent owners. Jarrell et al. (1988) confirmed this result: the market was working.

Jensen (1986) posited a mechanism by which value was being created: in an environment where the corporation had been taken private and the managers were general partners acting on behalf of limited partner equity investors, debt creation was a commitment device to limit the scope for losses due to agency problems. That is, by imposing debt on the corporation that was taken over (the target) and spending that debt on paying out the existing shareholders, management was committing to make substantial interest payments and thereby limiting its access to cash that could be redirected in the managers' own interests.<sup>3</sup> Jensen (1988) identified the issuance of high-yield bonds as a financial innovation that made the debt finance of these takeovers possible.

This literature overlooked two important factors: first, the buyout itself is founded on a conflict of interest as the beneficiaries of the debt that is being used to limit manager discretion are the new owners of the corporation, but they do not owe the debt since that obligation sits with the corporation that was purchased (the target); and, second, the majority of the debt that made leveraged buyouts possible was not market-based, but instead high-yield bank loans.

<sup>&</sup>lt;sup>3</sup> Lehn and Poulsen (1989) find empirical support for this result.

To clarify these points, an understanding of the term 'leveraged buyout' or LBO is essential.<sup>4</sup> An LBO is the purchase of a firm that trades on the stock market, 'taking it private,' but without paying cash for the shares. Instead the buyer/owner puts up about 10% of the price and borrows the remaining 90%. The catch is that the buyer/owner does not owe the debt, instead the target/purchased corporation owes the debt. In short, the private equity fund making the purchase gets all the advantages of ownership and control of the company, but has no liability whatsoever on the debt that was used to purchase the company.

Because the debtor is the corporation and the corporation does not get the benefit of the funds as they are fully paid out at the time of the transaction, it is not at all clear that these transactions should be legal – and they were very, very controversial when they initially began to be used. The presentation of the growth of these transactions as a 'market' phenomenon is also somewhat puzzling. In the 1980s between 50 and 90% of the debt that financed an LBO was provided by bank loans that were secured by the assets of the target company. The remainder of the debt was made up of junk bonds and/or other unsecured or subordinate financing. Thus, the privilege of borrowing at the scale that was necessary for these transactions required relationship-building with both the money center banks and the underwriters of junk bonds.

Furthermore, at their origin in the 1980s these transactions typically focused on large firms, so even 10% of the purchase price was a vast amount that far exceeded the private means of even the wealthy. Thus, the purchasers – leveraged buyout firms like KKR – sought out limited partner investors to fund the equity portion of the deals.<sup>5</sup> These funds then promised to have a lifespan of only seven to ten years and to give back the principal along with a healthy return at the end of the period. In short, far from eliminating the conflict between owner-principals and manager-agents, an LBO creates a new locus for the conflict in the fund that purchases the corporation.<sup>6</sup>

This paper uses call report data to demonstrate the significant role played by the largest US banks in the leveraged buyout boom of the 1980s and to explain why the large US bank chose to create the buyout boom. These banks were benefiting from extraordinary levels of foreign funding due both to 'petrodollars' and the US policy of too-big-to-fail, and built the syndicated loan market to facilitate the

<sup>&</sup>lt;sup>4</sup> While leveraged buyouts are just one of the mechanisms by which corporate structure is leveraged, the model of debt issuance provided by the leveraged buyouts of the 1980s has played an important role in the structuring of such transactions ever since.

<sup>&</sup>lt;sup>5</sup> Note that leverage buyout firms rebranded to private equity in the 1990s (Economist 2008; Kaplan & Stromberg 2009: 121).

<sup>&</sup>lt;sup>6</sup> This observation has been made by many others including Morris and Phalippou 2020: 308; Albertus & Denes 2023.

investment of these funds. Initially, the large US banks lent to low-income countries. However, when that lending went bad and the banks were treated to regulatory forbearance instead of being closed down, the large US banks needed another source of high-yield loans to rebuild their balance sheets: this was the leveraged buyout boom.

This finding that the LBO boom was bank-driven is important, because when bank lending is properly framed, it becomes clear that bank-driven lending can create economic distortions. We argue that the shift in the flow of US national income to the finance industry and the remarkable growth of US corporate debt can be considered two such distortions.

Section I discusses the theoretic framing used to discuss leveraged buyouts in the literature and proposes a reinterpretation of that framing. Section II presents the data on the role of banks in the leveraged buyout boom. Section III provides a history explaining why banks created the leveraged buyout boom in the 1980s. Section IV discusses how the bank-driven nature of the LBO boom created distortions in US national income and the market for corporate debt. Section V concludes.

# I. Debt and agency problems

An LBO is based on a very large issue of debt that is owed by the target corporation. The principal beneficiary of the debt is not the corporation itself, but the fund that uses the LBO to pay off the initial shareholders and take control of the corporation. Even so, the fund just like any other shareholder has no liability on the debt. A secondary beneficiary, as we will see, is the senior lender, who gets substantial fees from arranging the deal and also holds debt that pays a high yield of 200 to 300 basis points over the prime rate, while at the same time being protected by the assets of the corporation as collateral, such that if the loan goes into default the lender will have effective control over the corporation. The typical LBO funding structure in the 1980s is depicted in Figure 1.



Figure 1: Typical LBO funding structure (Borio 1990b: 7)

To frame the problem with this structure, first, one needs to recognize that a corporation's value is comprised of its assets, relationships, and streams of payment. These streams of payments include inflows from customers and outflows to employees, suppliers, creditors, tax authorities, and shareholders. These comprise the stakeholders in the corporation (see Dodd 1932; Freeman 1984). The knowledge embedded in the corporation provides it with investment opportunities that are not available to entities that do not have the corporation's detailed knowledge of the specific industries and markets in which the corporation operates, and Chandler (1977)'s history established the crucial role of manager-employees in building corporate value from the mid-19<sup>th</sup> century on.

The second step in framing the problem with the LBO structure is to recognize that historically large corporations have relied on external finance in the form of bank loans and bond issues to provide both working capital and investable funds (see Chandler 1977). Thus, debt issuing capacity has traditionally played an important part in the corporation's ability to invest and to build value for shareholders.

This traditional approach to the corporation recognizes that the conduct of managers is variable, that they will not always build value, and may indeed extract or erode the value of the corporation. Under the traditional theory the disciplining factor that can help prevent mismanagement is the public stock market.<sup>7</sup> Any investor in a public corporation who is dissatisfied with management can sell at any point in time – while the manager is committed to the firm, the shareholder is not. Indeed, if the shareholders exit *en masse*, the corporation's market value will collapse and the manager is likely to lose access to bank finance and with this loss of working capital the manager may well end up unemployed when the corporation enters bankruptcy.

In contrast to Chandler (1977)'s view of the corporation as the means by which manager-employees build value for shareholders, Jensen and Meckling (1976) reframe the corporation as a vehicle for *delivering* value to shareholders. Indeed, the corporation is reduced to a simple equation representing alternative options for managers, and no option for building wealth over time is modeled. Even though the stream of income that the corporation produces is entirely dependent on stakeholders and their continued participation in the corporate endeavor, by construct, the Jensen-Meckling framework treats the corporation's capacity to create value as independent of stakeholders. Thus, Jensen and Meckling effectively assume that the corporation exists only to the degree that it provides a stream of income for shareholders.

Jensen (1986) also reframes the value of cash flows to the corporation. While the traditional approach views borrowed cash flows as the means that make it possible for the corporation to invest and build value, Jensen (1986) focuses entirely on the conflicts of interest that managers may face and treats constraints on cash flows as benefiting corporate owners by limiting the decision space of managers. Indeed, because in this framework the corporation as an entity that can build value does not exist, the only question that is considered is how any current value in the corporation can be delivered to shareholders.

<sup>&</sup>lt;sup>7</sup> Jensen and Ruback (1983: 30-31) observe that in practice shareholders will have difficulty knowing when managers are acting in their own interests and as a result they will not know when to sell and will exert inadequate influence on self-interested managers. They treat the fact that LBOs were able to pay current shareholders more than the market price as evidence that the market price did not reflect 'underlying' value. The discussion in this paper indicates that the senior lenders' need for high yield revenues may have contributed to the high prices received by shareholders.

In reducing the challenge of corporate governance to a problem of managing the potential conflicts that may exist between shareowners and managers, Jensen and his co-authors effectively dismiss without discussion the substantial evidence set forth by Chandler (1977) that manager-employees produced dramatic benefits for both the American economy and the shareholders in the corporations despite these conflicts. They also overlook the fact that limitations on free cash flow in fact hobble the core function of the corporate form, which is to invest with a view to building value over the long-term.

In other words, imposing debt on the target corporation in order to make a massive payment to existing shareholders has the effect of limiting not just the managers' options, but also the corporation's options – some forms of value building and perhaps even maintenance will no longer be within reach. In the traditional framework, this would be understood as costly, but it cannot be costly in a framework where the corporation as a separate entity from shareholders does not exist.

When evaluating LBOs it is not, however, particularly logical to use a framework that assumes away the separate existence of the corporation. The reason this is illogical is that is that the existence of the corporation as a separate entity is fundamental to an LBO: it is the corporation that bears the debt in an LBO. Every other participant in the deal has limited liability.

In an LBO the senior creditor contributes the largest portion of the purchase price of the target corporation and demands the corporation's most important assets serve as collateral for the loan. The senior creditor is paid high yields on this debt and also typically receives fees for putting the deal together. The junior creditors contribute a smaller portion of the purchase price and are paid high yields for the risk of their investment using high yield bonds or similar obligations.

The purchasing fund pays only a fraction of the value of the corporation in exchange for control over the corporation and its assets. Because the loans that are owed to the senior and junior creditors are the obligations of the corporation, the buyer itself gets this controlling position even though the buyer has neither the resources needed to purchase the corporation nor the creditworthiness to borrow the money to do so. Furthermore, the purchasing fund replicates the principal agent problem that Jensen (1986) claimed that the LBO was designed to solve: manager-employees are replaced by manager-owner-agents who have a duty to represent the interests of their principals or the limited partners in the fund. All of the conflicts of interest embedded in the corporate ownership structure remain (Morris and Phalippou 2020: 308; Albertus & Denes 2023) – but at the cost of imposing on a formerly successful corporation a level of debt that will permanently hobble its capacity for building value in the future.

Furthermore, because the limited partner-principals do not have the ability to exit their investments by selling on a public market, the buyout fund is structured to have an end date at which assets will be sold

and returns will be paid out. This date is usually seven to ten years after the formation of the fund.<sup>8</sup> Thus, not only is the principal-agent conflict embedded in an LBO structure, but (i) the principal has less of an ability to monitor the behavior of the agent by using the public market price of the corporation as an indicator, and (ii) to offset this weakness a short-term investment horizon is imposed. In short, after an LBO, managers can no longer act in the long-term interests of the corporation (or the economy), but must work towards realizing value on a short-term horizon.<sup>9</sup>

In the LBO structure the senior creditor is effectively selling an option to the purchasing fund. In the event that the fund opts to have the corporation default on the debt, the senior creditor will effectively own the corporation.<sup>10</sup> The junior creditor protects the senior creditor from experiencing losses in the event that the option is exercised, because the junior creditor will bear losses before the senior creditor does.<sup>11</sup>

Overall, the effect of an LBO is to take a corporation that had been focused (perhaps imperfectly) on building long-term value for stakeholders and shareholders, and to turn it into a source of relatively short-horizon revenue for the finance industry. The imposition of a debt load on the corporation that is equal to 75% or more of the total current value of the corporation means all of the corporate stakeholders become part of a process of producing ongoing interest and dividend payments for the financial industry. Indeed, because of the high yields on the loans, these target corporations provide a much higher source of revenue for banks than they could possibly find elsewhere. The heavy burden of interest payments– deliberately – precludes the purchasing fund from having the resources to engage in significant investment in the future of the corporation, and thus the fund manages the corporation in order to draw value from it, not to build value in it.

In short, an LBO freezes the corporation's current value in place and delivers it as cash to the initial shareholders. The corporation then can no longer use its resources to build additional value over time, because the firm's borrowing capacity has been drawn down and the funds received have been distributed already. The corporation instead spends the rest of its existence in making payments to lenders and distributing value to the purchasing fund and any subsequent owners. In short, an LBO effectively sets off

<sup>&</sup>lt;sup>8</sup> While the general partner typically has some flexibility to extend the end date, this is very limited and unlikely to be for more than two years.

<sup>&</sup>lt;sup>9</sup> On the second point, see Christophers (2023).

<sup>&</sup>lt;sup>10</sup> In practice, in a bankruptcy scenario the senior creditor typically does not take control of the corporation, but instead dictates the terms of the reorganization plan or liquidation (see e.g. Ayotte & Morrison 2009; Lubben 2015).
<sup>11</sup> In the 1980s junk bond investors frequently comprised the junior creditors in these deals and at the end of the decade junk bond investors experienced losses – and there were prosecutions and convictions for the misselling of the bonds.

a prolonged liquidation in which the corporation's future prospects are absorbed in making payments to lenders and the shareholders.

Under these circumstances, it is not at all surprising that an LBO is experienced as a betrayal by stakeholders such as employees and customers (Applebaum and Batt 2014), as the prospect of a long-term value building relationship with the corporation has been eliminated in order to use the corporation as mechanism for producing revenues for the financial industry.

### A contemporary example

A contemporary example illustrates the problem. UK water companies were privatized in 1989 with the whole of their debt burden (£5 billion) written off by the government before the sale. Since that date their collective debt burden has grown to exceed £60 billion, and these private companies have not been engaging in the investments and maintenance required by law and as a result have been fouling waterways and the sea. The government is requiring them to address the dumping of sewage by spending £54 billion in capital investments over the next 25 years (Parker et al. 2023).

Thames Water is the company that serves London. It was purchased by Macquarie, a specialist in leveraged infrastructure investments, in 2007 at a price of £4.8 billion with borrowings of £2.8 billion. The terms of the acquisition explicitly required that the transaction not be structured as a leveraged buyout, and that £2 billion of the 'debt would be fully ring-fenced from Thames Water as the regulated entity' (Robinson 2017). Money is, however, fungible, and Macquarie was able to borrow against the value of the Thames Water assets in order to pay dividends that more than paid off the £2 billion in debt.

Over the period of Macquarie's ownership of Thames Water from 2007 to 2017, the company's debt rose from £3.4 billion to £10.8 billion, and Macquarie ultimately took out £2.7 billion in dividends and £2.2 billion in loans. Since 2017 Thames Water's debt has grown to £14.5 billion – and the holding company that owns the 'ring-fenced' water company owes an additional £1.35 billion, secured by the assets of the company (Plimmer and Fildes 2023). With interest rates rising, the company is likely to declare bankruptcy, long before the £1.5 billion committed as new equity by investors to address the infrastructure failures that have led to sewage spills is paid up. In short, Thames Water is a case where investors such as Macquarie have made high returns, at the expense of leaving urban Britain with grossly underinvested waterworks infrastructure. Someone will have to pay – and it is almost certain that the costs will fall either on ratepayers or taxpayers.

This example illustrates how the corporate form is not being used to raise funds for the purpose of providing valuable – and therefore profitable – services to the community, but instead to provide

9

investment returns to shareholders, with stakeholders such as customers treated as a stream of income from which as much value as possible should be extracted for investors. Similarly employee stakeholders are treated as cost centers who should be induced to work at the very limits of their capacity independent of the long-term effects on the employees. Instead of a win-win situation where surpluses are shared amongst stakeholders, these corporations seek to hold all other stakeholders to their reservation values.

Furthermore, these corporations also seek to extract as much value as possible from the government and society at large. Issuing secured debt ensures that when the corporation goes bankrupt it is the lenders, or the owners of the secured debt, who control the restructuring. Stakeholders such as employees, suppliers and customers with claims against the company are left holding the losses.

### II. Banks and the leveraged buyout boom of the 1980s: the data

As was noted above, in the 1980s the senior 50 to 60% of an LBO transaction was typically made up of bank loans (Thomson 1989; see also Borio 1990b: 7). Such bank loans are now called leveraged loans since they are made to highly leveraged companies. The bank holding companies for the largest banks also invested in the equity and mezzanine debt in these transactions helping to ensure that deals could be closed (Borio 1990b: 18; Thackray 1986).<sup>12</sup>

Contemporaries concluded that 'the flow of LBOs is created by lenders. They don't just support the market; they make it happen' (Thackray 1986). Indeed, the largest deals required the participation of the biggest banks that could take large portions of the loan to guarantee that they could be completed (Thackray 1986). In short, the too big to fail US banks were a core driving force behind the LBO wave (see also Seidman 1989: 4; Borio 1990b: 28; Osterberg 1993: 1).

The speed with which LBO lending developed was remarkable. Before 1980 'nobody knew what an LBO was. I mean nobody at the big investment banks – not just the cocktail party crowd' according to a partner in Forstmann Little, a firm that pioneered the strategy (Thackray 1986). By 1986 LBO deals accounted for more than \$30 billion every year and in 1989 \$50 billion (Borio 1990b: 9). Mergers and acquisitions more generally grew just as quickly – and while many M&A transactions do not result in an increase in corporate leverage, the LBO model meant that high yield senior debt became more common even in transactions that were not classified as LBOs. In each of 1986, 1988 and 1989, more than \$200 billion in M&A transactions took place (Borio 1990b: 9).

<sup>&</sup>lt;sup>12</sup> By 1989 the BHCs held \$9 billion in equity and mezzanine debt, which amounts to approximately 5% of the total issue of LBO equity and mezzanine debt issued up to this point in time.

By 1986 there were already complaints that the 'overhang of LBO-hungry money has revolutionized the [LBO] business' with the effect that LBO vehicles were being used for an ever broader range of industries. In 1988, the Chairman of the FDIC reported that 10% of the 'big banks' new business loans are for buyouts' (Knight 1988). This situation of vast financing flows chasing deals was driving such aggressive pricing that it was clear that it would be difficult for the underlying cash flows to support the debt (Thackray 1986).

To substantiate this narrative the US call reports have data from 1984 through 1993 on 'Loans originated by the reporting bank that have been sold or participated to others during the calendar quarter ending with the report date.' LBO loans were so large that the senior creditors would form a syndicate to originate the loan, and then would sell participations in the loan to others. Chart 1 displays the quarterly data on loans sold/participated for all call reporting institutions as a percent of commercial and industrial loans held at the end of the quarter. It is remarkable that the number of loans reported as sold/participated in a single quarter rises in 1989 to almost 40% of all commercial and industrial loans held on the balance sheets of all the banks in the banking system. The subsequent decline is more easily explained, as many of these loans went into default generating lawsuits and some bank failures.



When one drills down into the call report data, the numbers of loans sold/participated are startling. Chart 2 sums across the quarterly data to generate annual figures for the number of loans sold/participated. In 1988 and 1989 the sum across all reporters exceeds \$1 trillion – at a time when total assets in the banking system were \$4 trillion. It is almost certain that this figure must represent some double-counting where two or more banks were originators of a syndicated loan and both report the loan as sold/participated.

To get a better understanding of this phenomenon Chart 2 also depicts the figures reported by the 11 largest banks in 1984 – which are the banks that reported the highest number of loans sold/participated across the decade.<sup>13</sup> While the largest banks accounted for just 38% of loans sold/participated in 1984, by 1985 they accounted for more than 60%. This rose to 86% in 1988 and dropped back down to 70% in 1993.



To address the double counting problem, we make the conservative assumption of triple reporting across the 11 largest banks and divide the loans that they reported as sold/participated by three. Based on this adjustment in 1988 and 1989 the 11 largest banks accounted for \$300 billion in loans sold/participated and in 1987 and 1990 for \$200 billion.

From this we can draw three conclusions: First, over a period of just four years from 1984 to 1988 a stunning shift in the lending behavior of the largest US banks took place. Second, the footprint of the largest banks in US commercial and industrial lending changed dramatically over the course of the 1980s.

<sup>&</sup>lt;sup>13</sup> We include 11 banks because Mellon is the eleventh largest bank, and as the chart indicates played a role in this lending. On the other hand, the 12<sup>th</sup> largest bank, Wells Fargo, was not involved.

As Chart 3 shows the conservative one-third measure of large bank loan sales/participations accounted for as much as 40% of all commercial and industrial loans on the balance sheets of US banks at the peak.<sup>14</sup> Third, the shift in the structure of US bank lending persisted into the future. Chart 3 shows that even after the rush of loan sales/participations declined, the conservative estimate of this segment of the 11 largest banks' lending still amounted in each year to almost 13% of all commercial and industrial loans in the banking system.



While this transformation in the lending behavior of the largest banks was not limited to financing LBOs,<sup>15</sup> observers of the phenomenon identify mergers and acquisitions as a key driver of this lending, and also as a factor that explains the increase in the margins earned by the banks on lending at this time (see e.g. Allen 1990: 73).

<sup>&</sup>lt;sup>14</sup> Of course, not all of these loans were sold to US banks (see Allen 1990).

<sup>&</sup>lt;sup>15</sup> Security Pacific National Bank, in particular, which reported the highest total quantity of loans participated/sold from 1984-93, was selling off a very wide variety of loans and by 1990 was a troubled bank that would ultimately be purchased by Bank of America. Some information on Security Pacific NB's loans is available in lawsuits such as Banco Espanol de Credito v Security Pacific, 973 F.2d 51 (2d Cir. 1992).

Of course, the large banks did not just sell and participate these loans – they also held them on balance sheet. Seidman (1988: 4) reports that the 10 banks most involved in LBOs held \$20.2 billion in LBO loans on their balance sheet on 30 September 1988. While Seidman does not identify the 10 banks in question, by summing across the C&I loans at the 10 largest banks in Chart 2,<sup>16</sup> we find that the 10 banks had \$156.8 billion in C&I loans on balance sheet on 30 September, and thus that about 13% of their C&I loans were LBO loans. Of course, some banks held more LBO loans on balance sheet than others: in 1988 almost 25% of Bankers Trust's C&I loans were LBO loans.<sup>17</sup> Consistent with what we have found above, Seidman (1988: 4) reports that the big banks typically held only 10% or less of a transaction and would sell participations in the remainder. While Seidman (1988: 4), the FDIC Chairman at the time, was attentive to the risks of LBO loans, he also makes it clear that the market is very advantageous for the big banks, because it provides them with yields that are typically about 200 basis points above the yield on conventional commercial lending (Borio 1990b: 22). This high yield is in addition to the fees earned from origination and sales of the loans. Initially regulators appear to have largely approved of LBO lending because of the high earnings it provided to banks together with confidence that the risks to the banks could be managed (see e.g. Thomson 1989: 4; Seidman 1989: 4).

Banks such as Security Pacific were however in trouble by 1990, and Osterberg (1993: 1) observes that 'the aftermath of the wave of leveraged buyouts in the 1980s brought a realization that banks had played a key role in financing such transactions.' This led regulators to require detailed reporting on highly leveraged transactions from 1991. The regulators, however, withdrew the requirement for such reporting after only six quarters, finding that the regulated entities had already improved their practices and that the reporting 'may be having an undue effect on pricing and availability of credit to certain highly-leveraged borrowers' (Fed Board 1992). Currently, the data that was reported in 1991 and 1992 is not in the call reports that are available publicly. Overall, the bank regulators were apparently willing to support the growth of this bank lending activity, because it was so profitable for the big banks.

### III. Bank lending in the post-Bretton Woods era

To understand why the largest US banks drove the growth of LBOs, one must recall the complexity of the economic environment in the early 1980s. To set the stage we start a decade earlier in the 1970s.

<sup>&</sup>lt;sup>16</sup> That is, we exclude Mellon as the smallest bank from this calculation.

<sup>&</sup>lt;sup>17</sup> LA Times (1988) reports that Bankers Trust had a \$2.7 billion LBO portfolio. On 31 December 1988, Bankers Trust reported \$10.95 billion in C&I loans.

On August 15, 1971 President Nixon suspended all convertibility of the US dollar into gold, setting the stage for the 1973 shift to an international system based on floating exchange rates. At this time, it was far from sure that the dollar could retain its international role, and efforts were made to denominate Eurocurrency lending in a unit linked to a bundle of European currencies (Rothschilds 1971; Bank of England 1974). Indeed, in 1978 the UK Prime Minister consulted with the Bank of England Governor and other experts about the possibility of promoting an alternative to the dollar in denominating international debts and was disappointed – and not entirely convinced – by the finding that, given the current market share of less than 5% for 'units of account' such as the SDR, this possibility was deemed impractical (Bank of England 1978). In fact, from 1971 through 1973 Eurobonds were increasingly denominated in currencies other than the dollar, which saw its share fall from 49% to 37% (Bank of England 1974). The Deutschemark and the Swiss franc were the principal competitors.

In short, the early-1970s were years when it was possible that international financial markets would transition away from reliance on the dollar. Both Germany and Switzerland sought, however, to limit Eurocurrency issues in the Deutschemark or franc respectively by adjusting controls on the market (Allen 1971, BoE 1973), whereas the US authorities were actively supportive of the use of the US dollar in international markets.

In this environment, the treatment of a US bank that was both poorly managed and active in international debt markets, Franklin National Bank, is instructive. Franklin National was a New York bank that began to grow quickly in the 1960s, and opened a London branch in 1972. Participation in the international Eurodollar markets that were centered in London was attractive, because this avoided domestic US regulation and thus offered the opportunity for lower cost funding and lending activities. By 1973 Franklin National was the 20<sup>th</sup> largest bank in the US. It had also for several years been a 'problem bank' meaning that its regulator had concerns about its capital and lending practices (Spero 1980: 49-50). Despite this the US regulator asked the Bank of England to give the London branch full recognition in November 1972.<sup>18</sup> Franklin National promptly set off to improve its weak earnings performance by speculating on foreign exchange markets. Within a year, representatives from Morgan Guaranty Trust, a prominent US bank, were meeting with the president of the New York Federal Reserve Bank to express concerns about Franklin's activities and report that Morgan and other banks in London were refusing to enter into certain transactions with the bank (Spero 1980: 66). By May 1974 Franklin had lost access to Eurodollar funding entirely.

<sup>&</sup>lt;sup>18</sup> It was standard practice for the Bank of England to seek approval from domestic regulators before authorizing a London branch.

The Federal Reserve deemed that a failure would jeopardize financial stability 'with further serious repercussions for domestic and international financial markets in general', and as no merger partner could be found to take on Franklin's losses, began to fund Franklin National through the discount window (Spero 1980: 122-25). Within two months the Federal Reserve was lending Franklin \$1.2 billion, more than a quarter of Franklin's balance sheet (Spero 1980: 127). This lending continued – and grew by almost 50% – until October when a buyer was found for a selection of 'good' assets. The Fed had to buy Franklin's foreign exchange book and operate it in runoff, indemnified by the FDIC which took the remainder of the assets into receivership.

To summarize, in 1974 the resources of the Federal Reserve and the US government were used to ensure that a demonstrably bad bank (whose officers would later be convicted of fraud) did not impose costs on international financial market participants. This contrasts sharply with the June 1974 failure of Bankhaus I.D. Herstatt, a German bank that was also engaged in currency speculation on international markets. Not only did the German government refuse to support the bank, but its closure while markets were open was so disruptive, we still speak of 'Herstatt risk.'

In short, from 1974 the US authorities made it clear that they stood ready to protect the Eurodollar market from US bank failures. After this a Fed governor would describe US banking as a 'no failure industry,' and the Fed chairman acknowledged the extraordinary role played by the Fed and asked how to protect 'the free enterprise system' in the US (Coldwell 1976; Burns 1974). The predominance of the dollar in Eurocurrency markets was not a 'market' phenomenon, but US government policy.



To help understand these policy decisions Chart 4 presents the foreign interest expense of the 10 largest US banks (which accounted for about 25% of the total assets of all call reporters).<sup>19</sup> The series that is available from 1976 is the interest expense paid on foreign deposits as a fraction of total expense. By 1976 foreign interest expense was already 35% of total expenses. By 1980 this figure had risen to almost 50%. This model was particular to the largest US banks. In 1976 the top ten banks accounted for 78% of the interest paid on foreign deposits by all call reporters. While this ratio declined fairly steadily to 69% in 1989, it rose afterwards and has remained higher ever since.

To interpret this data, it is important to recall that domestic demand deposits in the US were not interestbearing in this period. Demand deposits were also, however, a declining fraction of the US deposit base. FDIC data indicates that interest-bearing deposits in the aggregate commercial banking system increased from 50% of deposits in 1976 to 71% in 1980, rising to 82% by 1990.<sup>20</sup> Overall, the interest expense data

<sup>&</sup>lt;sup>19</sup> Note that the call reports do not provide a separate category for foreign deposits in this period.

<sup>&</sup>lt;sup>20</sup> Data downloaded from <u>https://banks.data.fdic.gov/explore/historical</u> in 2019.

makes it clear that foreign funding was a very important component of large bank funding in the 1970s and early 1980s.

The series that are available from 1984 on support this conclusion. The interest expense paid on foreign deposits held steady at about 45% of total interest expense until the mid-1990s, while the interest expense booked abroad rose in this period to exceed 75% of total interest expense.

Why were the largest US banks drawing so much of their funding from overseas? Part of the explanation is that the banks were not subject to US bank regulation on their foreign funding and so it was less costly. Also important, the first oil price shock took place in 1973 and create large dollar balances held abroad by the OPEC member countries. The role played by oil prices is made clear by the sharp rise in interest paid on foreign deposits from 39% of expenses in 1978 to 45% of expenses in 1979, when the second oil price shock took place (see Chart 4).

This heavy reliance on foreign funding was almost certainly both a cause of the US policy of protecting international markets from losses due to a US bank failure, and also a consequence of this policy: it gave US banks an advantage in raising funds on these markets. Overall, US policy in this era created a feedback loop that promoted large bank reliance on foreign funding.

The 1980s saw additional bank bailouts. Indeed, the 'too-big-to-fail' policy that was a concern for bank regulators after 1974, became publicly acknowledged in 1984. In 1980, the Fed repeated its 1974 behavior, extending and rolling over large loans to a failing bank, First Pennsylvania, until such time as the FDIC was convinced to bail it out (Sprague 1986: Ch 5). 1984 saw the most famous example of a 'too big to fail' bank, Continental Illinois, which was taken into FDIC ownership, but allowed to continue operating until it was finally sold. This was the culmination of a decade in which the Fed had made it clear that if a bank was active in the Eurodollar market, it would not be allowed to impose losses on creditors.

Oveall, US policy in this period both provided very strong support for the dollar in Eurocurrency markets and fostered the growth within the US of a two-tier banking system comprised on the one hand of those banks that were active in international markets, and thus could not be allowed to fail, and on the other hand those that were smaller and domestically-focused – and subject to be put into FDIC receivership when no merger was possible. The more important issue, however, is what the banks deemed too big to fail would do with the vast funding flows that were created by US policy.

## Investing 'petrodollar' inflows

In the 1970s and 1980s the too big to fail US banks accepted a large inflow of funding that was not associated with US commercial activity. In order to pay the interest due on that funding, they had to invest in earning assets. To make use of the sizable funding inflows, the banks formed syndicates that made large loans – in the tens of millions of dollars – and diversified their lending by taking an allocation of only a portion of the loan, and also by selling off participations in the loan after it had been made.<sup>21</sup> US banks took a dominant position in international syndicated lending in the early 1970s (Bond 1985: 16).

Who were the borrowers who could qualify for such large loans? In the 1970s the borrowers were typically sovereign debtors or businesses and projects that were guaranteed by sovereigns.<sup>22</sup> When the less developed country debt crisis revealed the dangers of lending into the high yield segment of the sovereign market, leveraged buyout funds and other purchasers of large corporations filled the gap, as we have seen above.

### Foreign lending

Even before the first oil price shock of 1973, concerns were being raised in London about the growth of Eurocurrency lending to developing countries, which was estimated to have quadrupled from 1971 to 1973 (Wright 1973; BoE 1973b). This growth was driven by a combination of less onerous lending criteria than traditional bilateral lending, administrative simplicity, and abundant liquidity on Eurocurrency markets as more and more banks joined and became lenders on the market. The fact that these were term loans of five to ten years with floating, and not the traditional fixed, rates was immediately flagged as a potential concern (IFC 1973: 4-5).

Concerns about the rapid growth of eurocurrency loans to developing countries culminated in a December 1973 consultation with private-bankers (BoE 1973b). The bankers predicted that this lending was likely to grow due to the anticipated inflow into Eurocurrency markets of funds from oil exporting countries. Lending did grow through the first quarter of 1974, but the growth was temporarily disrupted by the failure of the Herstatt bank, resulting in increased rates and shorter loan-terms along with a drop in total lending back down to the 1973 level for 18 months.<sup>23</sup> In 1976 however the market resumed its rapid growth, doubling in size from 1977 to 1978. The growth in lending to low-income countries was driven

<sup>&</sup>lt;sup>21</sup> This technique had originally been developed for corporate lending in the US in the 1950s after the IRS approved an accelerated alternative to straight-line depreciation for assets (Bond 1985: 15).

<sup>&</sup>lt;sup>22</sup> 'Only governments can give assurances on interest and loan repayments because it is they who control the foreign exchange reserves of their countries.' Indeed, loans to finance business projects would be evaluated with less strict criteria in the presence of a government guarantee (BoE 1973b).

<sup>&</sup>lt;sup>23</sup> After the Herstatt failure smaller banks were shut out of London interbank market entirely for 4-6 weeks. Slow recovery to normal meant that spreads on Eurocurrency markets continued to rise through mid-1975. Banks became cautious and were very careful not to over-extend themselves through 1975 (Bond 1985: 20).

not just by the liquidity pouring into the market, but also by the fact that many countries were running short of dollars and could access dollar liquidity through the Eurocurrency market (Bond 1985: 21). By the end of 1978 spreads for low-income country borrowing had fallen back below 1%. While the growth of the market stagnated for the next few years, the absolute amount of lending continued to be high at about \$80 billion per annum.

Concerns about bank lending to low-income countries was so high that Japan in October 1979 effectively blocked its banks from participating in these loans. Bond (1985), a Bank of England economist, reported that, since the interest of small banks in participating in the international market was also drying up, in the early 80s it was frequently the largest banks that were taking up all of a loan. In the US, from 1979 bank regulators were strengthening capital requirements and using supervisory authority to draw attention to loans to highly indebted countries. These efforts had however little effect on the growth of aggregate bank lending (FDIC 1997: 89, 203; Wallich 1981: 13). Even though the share of foreign sovereign loans held by the smaller US bank increased dramatically from 1979 through 1982 (see Chart 6), by the end of 1981 the exposure of the largest US banks to developing country debt comprised more than 10% of their assets and 2.6 times their capital (FDIC 1997: 196, 199; Boughton 2001: 283-86). Overall, despite their efforts US bank regulators demonstrated that they had less ability to constrain bank lending than regulators in other countries.



Call report data shows that by 1978 almost 60% of the commercial and industrial lending by the 10 largest US banks was to non-US addresses, and these 10 banks accounted for 83% of the C&I lending to non-US addresses of all call reporters.<sup>24</sup> This foreign commercial lending comprised 17% of the 10 banks' total assets, whereas lending to foreign sovereigns was over 4% of total assets (see Chart 5). The lending of these 10 banks to foreign sovereigns was 73% of the lending of all call reporters to foreign sovereigns (see Chart 6).

Even though the foreign loans held on balance sheet by the 10 largest banks increased from 1979 to 1982, their share of foreign lending by US banks dropped dramatically from 83% to 68% for C&I lending and from 73% to 46% for sovereign lending (see Chart 6). Over these three years the 10 largest banks' non-domestic C&I loans increased by 32% to \$108 billion and their sovereign lending by 17% to \$21.5 billion. Lending by all call reporters to non-domestic C&I borrowers increased over these three years by 60% to \$159 billion and to foreign sovereigns by 83% to \$46.6 billion.

<sup>&</sup>lt;sup>24</sup> In 1978 the C&I lending to non-US addresses of all reporters was \$92 billion of which \$76 billion was on the balance sheets of the top 10 banks.



Note that the apparent explanation for the increase in fraction of C&I lending by the 10 largest banks from 1977 to 1978 is that the banks only started to report non-domestic C&I lending in 1978.

Given this data, differences in the regulation of banks in England and the US are of interest. An exchange within the Bank of England lays out a range of views about the market in London. A member of the Overseas Office distinguishes between "responsible members of the London market" who will avoid tempting a borrower into over-extending and "some of the banks seeking to gain an entry into this market (this means Japanese and newly-arrived US institutions) [who] are too willing to lend, at excessively low margins, and with inadequate examination of the borrower's prospects." (Edgley 1973). A colleague concurs: "the City is largely self-policing" because it is populated by banks that "are dependent on the esteem in which they are held by others for business" and the "constant survey, and assessment, of other banks' actions, … seems to work, for Euro-currency credit losses in recent years have not been of London's making." He continues: "I do think it rather stupid of people to suggest that London bankers are really so gullible as to allow themselves to become the dustbin for a mass of lending which the odds suggest will not be repaid." And concludes: "No doubt authorities abroad are as concerned about the

position of their country's banks and look at their overall business, which would include any London branch" (Galpin 1973). In short, UK regulators were confident that the incentives faced by UK banks would ensure that they did not engage in lending which would not be repaid.

Contemporary US regulators had no such confidence about US bank lending to low-income countries. Fed Chairmen Arthur Burns and Paul Volcker both correctly predicted that US banks would "extend credit more generously than is prudent" with significant implications for the vulnerability of the international credit structure, implicitly acknowledging that the incentive structure faced by US banks was broken (Burns 1977; Volcker 1980: 21, 27, 31-32). Volcker makes the key point that this debt served to maintain strong export markets for the US and other developed countries in difficult economic circumstances – and this fact can explain the support of other government officials for the continued growth of LDC lending (Kapstein 1994: 85, quoting Deputy Secretary of State Elinor Constable).

Mid-1982 was the turning point in syndicated lending to low-income countries, which were faced with prolonged high interest rates raising the debt service on their Eurocurrency debt together with a slowdown in global growth that reduced their ability to raise export-driven income to service the debt (Bond 1985; Broughton 2001: 283). After the Mexican suspension of payments, this lending dried up.

At least seven of the 10 largest banks in the US would have failed if they had been forced to recognize losses on lending to low-income countries promptly (FDIC 1997: 207). Instead they were protected from failure by a combination of regulatory forbearance and concerted efforts by US government officials in tandem with the IMF and the biggest banks to induce Latin American borrowers to avoid outright default by extending the debt. This episode has been well documented elsewhere and will not be discussed further here (see e.g. Kapstein 1994; FDIC 1997; Boughton 2001; Torre 2021).



(Note this chart can be expanded to include all quarterly data rather than just fourth quarter data.)

To understand how very troubled the largest US banks were at this time, Chart 7 shows the write-offs due to foreign lending that these 10 banks had to take in the last quarter of each year as a percent of assets. From the start of the series in 1984 through 1992, the largest banks were recording significant losses. Chart 8 makes clear how these losses affected the largest banks. In 1979 their assets accounted for 27% of call reporter assets. By 1991 they only accounted for 16.5% of call reporter assets. From 1982 to 1991 the total assets of the 10 largest banks grew by only 16% due to all the losses they had to absorb. By contrast the banking system as a whole grew by 69%.



To summarize, just prior to the LBO boom, most of the 10 largest US banks were insolvent, but they continued to receive significant funding on international markets – in reliance on the too big to fail backstop from US authorities (see Chart 4).<sup>25</sup> After the big US banks were protected from realizing their losses on lending to low-income countries, they still had abundant funds to allocate and a well-developed syndicated loan market to help them do so. Furthermore, the syndicated loan market was 'one of the few sources of high-profit lending' for the large banks due to fee income: the managing bank(s) in a syndicate would earn a fee of 1% of the transaction and would earn another 0.75% on the part of the loan that each bank took up (Thackray 1986; see also Bond 1985: 55-56). This of course was in addition to the interest paid on the loan. In short, what the banks needed was borrowers for their syndicated loans that could help them earn their way out of the deep hole they were in and that regulators would not object to.

It is therefore not at all surprising that a new syndicated lending market, the leveraged loan market, exploded in the US in 1983 (Knafo and Dutta 2016; see also Thackray 1986),<sup>26</sup> accompanied by a great deal of critical commentary (Shad 1984; Thackray 1986).<sup>27</sup> Indeed, the close connection between the

<sup>&</sup>lt;sup>25</sup> Note that the Eurocurrency market remained unregulated. US regulators had sought in 1979 to impose reserve requirements on Eurocurrency deposits, but were unsuccessful in coordinating with foreign regulators to do so (Hawley 1984: 152-59).

<sup>&</sup>lt;sup>26</sup> Allen (1990: 72) identifies 1986 as the year in which the international segment of this bank lending began to grow.
<sup>27</sup> In 1984 Karen Horn made the following comment at an FOMC meeting: 'Another Fourth District story about banks ... around the country that I found very interesting was in connection with this recent takeover bid by

collapse of LDC lending and the growth of the leveraged loans that finance LBOs was clear to contemporary financial market participants (Thackray 1986; see also Altunbas et al. 2006).<sup>28</sup> Thus, the data presented in Section II is explained by the largest banks' need to develop a relatively safe, but high-yield asset to help them earn their way out of their difficulties.<sup>29</sup> This also explains why regulators saw this form of lending as benign – it was helping the banks out of their difficulties and efforts to close this form of lending down were likely to result in more bank failures. When call reporting on highly leveraged transactions affected the market for them in 1992 the regulators stopped the reporting after only six quarters (see Section II).

While the evidence is strong that it was syndicated loans from the banks that explain the fact that the LBO market exploded in 1983, it is important to observe that both the junk bond market (which provided subordinated finance) and the limited partnership investment fund (or more specifically the LBO fund which provided equity for the transaction) had to be developed before the buyout boom could take place. The development of the junk bond market has been studied in great detail by others (e.g. Knafo & Dutta 2016; Knafo & Dutta 2019; Lazonick & O'Sullivan).

The growth of leveraged buyout funds was made possible by reform of the regulations governing investment funds. The Security and Exchange Commission's Regulation D was a response to the mandate given the SEC in the Small Business Investment Act of 1980 changing its focus from being strictly limited to investor protection to include the goal of facilitating small business capital formation too. This law also created the concept of the "accredited investor," explicitly including pension funds in the definition along with other institutional investors. The SEC promulgated Regulation D in March 1982, removing strict limitations on what could be considered a "private offering" and therefore exempt from the formal securities registration process that was required before marketing an investment to the public. Prior to 1980 the exemptions were very narrow, limited by either the number of investors (35) or the amount raised (\$100,000).

the fellow who controls that, needed--I don't remember the exact number--in excess of a billion dollars to make the bid. And for reasons of confidentiality, he didn't want to have a bank form a syndicate to get the loan for him. So, he did what he was advised couldn't be done. He went to thirty-two banks and asked them for lines of credit for a takeover for an acquisition and he wouldn't tell them the acquisition target, and out of the thirty-two very large banks he approached only one refused to give him a line of credit because they wanted to know the name of who was being taken over. Then, on a Friday afternoon, within several hours he pulled down all the lines and made the offer. It is, I think, an indication of how business is being done these days.' (Horn 1984: 9).

<sup>&</sup>lt;sup>28</sup> This market had the added advantage that banks could also earn fees on hedges for the LBO's floating rate debt (Thakray 1986).

<sup>&</sup>lt;sup>29</sup> Shleifer and Vishny (1990) emphasize how 'the easy availability funds' made the takeover wave of the 1980s possible – but don't explain how this is possible given that throughout the 1980s interest rates remained remarkably high after the Volcker shock of the early 80s and indeed the bank prime rate only fell below 7.5% at the very end of 1991.

Regulation D was organized around the concept of the "accredited investor," stating explicitly that those who fell within this definition did not require the investor protections created by the Securities Act of 1933, and expanding the definition to include wealthy individuals. Reg D created Rule 506, a new exemption for issues that did not involve "general solicitation" or in other words advertising to the general public.<sup>30</sup> The Rule 506 exemption was for issues in an unlimited amount to an unlimited number of accredited investors and up to 35 additional, sophisticated investors.

Within 4 years, newspapers were reporting 'The appetites of US investors ... seem insatiable. There is probably around \$5 billion committed at present to equity partnerships' (Thackray 1986). By 1989 these funds could raise \$14 billion in commitments in a single year (Fenn et al. 1995: 12). Only after the SEC's Rule 506 was promulgated was it possible for leveraged buyout funds to raise this kind of money.<sup>31</sup>

While all three forms of financing, equity (provided by leveraged buyout funds), mezzanine (often provided by junk bonds), and senior debt (provided by syndicated bank loans), played an important role in LBOs, as Figure 1 shows, bank loans provided most of the financing for LBOs: it was only possible for leveraged buyout transactions to be so large because the banks were seeking to lend vast sums.

As was predicted in 1986, by the end of the 1980s the leveraged buyout market was in trouble. In 1989 the junk bond market crashed (and both Drexel and Michael Milken had been charged with market manipulation). These losses – on top of the losses from lending to low income countries – and additional losses in real estate left two banks in precarious condition. Citibank and Bank of America were under-capitalized and had to be protected by regulatory forbearance granting them time to rebuild their balance sheets and recover their losses (Fromson and Knight 1993; Wilmarth 2002: 305; Wilmarth 2014: 76-77). (The Federal Reserves' low interest rates of the early 1990s also helped.)

At this time Citibank was described as having a culture that 'hunger[s] for revenue rather than profit' (Lee 1992). In short, the behavior that had been normalized amongst the more aggressive US too-big-to-fail banks was precisely the 'gullible' behavior that a Bank of England official had found it would be 'stupid' to imagine in English banks. Another way of describing this behavior is that too-big-to-fail banks, because they knew the government would act to bail them out, were making non-economic loans and losing money on a regular basis due to moral hazard. Even in the earliest days of the leveraged loan

<sup>&</sup>lt;sup>30</sup> Note that the JOBS Act of 2012 relaxed the general solicitation requirement.

<sup>&</sup>lt;sup>31</sup> Note that there were additional legal changes of the late 1970s and early 1980s that were necessary to make it possible for pension funds to invest in limited partnership investment funds (Fenn et al. 1995: 10-11). The role of consulting companies such as McKinsey and Bain in the LBO boom also merits study – Bain Capital was after all spun off of the Bain consulting company in 1984 and would regularly hire its parent company to advise its portfolio companies. McKinsey advised banks on strategy in the 1980s (Hayes 1983), LBO firms were prominent McKinsey clients (Acharya et al. 2013), and McKinsey produced glowing reports on LBOs (e.g. Butler 2001).

market, government subsidies of too big to fail banks played an important role in their growth, in the LBOs that they financed, and in the remarkable returns earned by leveraged buyout funds.

#### *Rebutting the standard narrative*

The academic literature describing the sudden onset of the buyout boom and shifting portfolios of the largest banks typically describes the banks as responding to a difficult competitive environment, and remarkably, rarely mentions the fact that the losses created by the overhang of bad loans to low-income countries were what made the 1980s environment challenging for the largest banks. Instead, deregulation is treated as having exposed banks to new 'market-based' competitors (e.g. CBO 1994; Borio 1990a: 37). On the lending side, the list of bank alternatives typically includes sources of borrowing for corporations such as commercial paper, junk bonds, and international banks. On the funding side, money market funds are the most important competitors.

These consequences of deregulation provide a good description of what made the late 20<sup>th</sup> century environment inhospitable for smaller banks: they had to compete for funding with interest-paying competitors like money market funds and were unable to keep any large corporate clients as borrowers, because they now had access to a wide variety of funding options. It is, however, an extremely poor description of the effects of deregulation on the environment faced by large banks.

First of all, deregulation is what allowed the large banks to be active on Eurodollar markets, giving them a cheaper source of funds than the small banks. After the 1974 bailout of Franklin National the biggest banks had an additional 'too-big-to-fail' boost from the US regulators to their ability to raise funds on international markets. The largest banks had government-supported credit ratings (see e.g. Moodys 2010) allowing them to borrow easily on commercial paper markets, in addition to earning fees from providing standby facilities to corporate commercial paper issuers. Thus, it is hard to understand how the deregulation of the 1960s, '70s and '80s can be considered a source of more difficult funding for the largest US banks.

Arguments that deregulation forced banks to adapt to a world where borrowers could rely on junk bonds or foreign banks similarly fail to make the crucial distinction between the challenges faced by smaller banks and the significant advantages that deregulation provided to the too-big-to-fail banks. As was discussed above the leveraged lending of the too-big-to-fail banks helped supercharge the junk bond market, because junk bonds provided a layer of subordination to make the bank loans senior and safer. Thus, the junk bond and leveraged loan markets grew symbiotically. Furthermore, while it is true that international banks expanded more deeply into US markets at this time, they did not penetrate US market more deeply than the too-big-to-fail US banks penetrated foreign markets. And when it came to

28

syndicated lending, both to low income countries and to buyout firms, there is no question that the US banks were the leaders of both phenomena.

Finally, although the largest banks did face challenges through the 1980s and were growing much more slowly than the smaller banks, this can be accounted for almost entirely by the writeoffs that the banks had to take over the course of this decade on their international loans. In short, their troubles in the 1980s were of their own making. We can conclude that the too-big-to-fail banks grew the leveraged lending market, not because competitive pressures forced them too, but because they needed a source of high-interest loans and fees in order to offset the writeoffs that they were having to take on their bad loans to low-income countries.

## IV. Does it matter if LBOs were a bank-driven phenomenon?

We have shown that the wave of LBOs that took place in the 1980s was a bank-driven and not a marketdriven phenomenon. To understand why this matters, we first explain 'banking theory,' a 19<sup>th</sup> century approach to banking that views bank lending as having the capacity to generate massive distortions in the economy including asset price booms and busts, and that shaped the development of central banking (Sissoko 2016; Sissoko 2022; Sissoko forthcoming). Then we discuss the distortions that are created by the bank finance of private equity.

### A framework explaining why bank lending controls are essential to avoid economic distortions

Banks in the modern period developed to address a specific type of financing need: it is inefficient for a firm to be financially constrained such that the only reason its production is limited is insufficient funds to purchase the inputs that are needed to complete profitable sales a few months in the future (Smith 1776: II,ii).<sup>32</sup> Historically banks addressed this problem in a very cost-effective manner by monetizing short-term credit in the form of bills of exchange (Smith 1776: II,ii; Sissoko 2016; Sissoko 2022).<sup>33</sup> Notably, this development of banking took place in an environment where the usury rate was capped at 5% per annum.

The availability of inexpensive bank-based short-term credit, however, very quickly raised the issue that when bank finance is too readily available, such that a business can continuously borrow funds to maintain its operations, the business is protected from going bankrupt, even if its business model is in fact deficient and sales are not profitable (Smith 1776: II,ii).<sup>34</sup> Thus, the key question for nascent central

<sup>&</sup>lt;sup>32</sup> "The commerce of Scotland, which at present is not very great ..." and following paragraphs.

<sup>&</sup>lt;sup>33</sup> The system of banking based on deposits that fund loans evolved from this system of bank-supported bills, and indeed the check is legally even now just a bill drawn on a bank (Sissoko 2022).

<sup>&</sup>lt;sup>34</sup> "The operations of this bank seem to have produced effects quite opposite ..." ff

bankers was the question of credit control: how to ensure that banks provide adequate credit for the economy to grow without providing so much credit that ultimately an excess of business failures would destabilize the economy (Monnet 2018).<sup>35</sup> One important principle was that banks should avoid investing in business equity, and should focus on making carefully monitored short to medium-term loans. Part of the monitoring process was to ensure that the borrower had the means to pay back the debt and that the roll-over of funding that paid back both the interest and principal of maturing debt was not serving to hide a *de facto* equity investment by the bank in the business. Another important principle was that in order to avoid asset price bubbles bank lending that was collateralized by long-term assets had to have its growth strictly limited by criteria other than the price of the underlying assets (Sissoko 2016).<sup>36</sup>

Since that time, the problem of constraining the growth of bank lending to prevent asset price bubbles and financial instability has been an ongoing theme of financial regulation (Sissoko 2017; Schumpeter 1939; US Senate 1933. See also Bindseil 2019; Monnet 2018; Ugolini 2017). Some of the principles underlying bank-driven financial instability have been rediscovered by modern scholars: Adrian and Shin (2010) and Brunnermeier and Pedersen (2009) show that, when bank lending is based on collateral, feedback loops and asset price bubbles can arise,<sup>37</sup> and Sissoko (2021) models the role of banks in the payments system and concludes that financial stability requires policymakers to regulate competition in the banking sector.

The growth of leveraged loans can be viewed from this perspective, treating banks and their credit origination practices as a key determinant of both economic growth and financial instability. From this viewpoint, the growth of leveraged loans, both in the early years and in the following decades, raises immediate concerns that bank credit may be facilitating the on-going operation of corporations without strong fundamentals or that it may be having an adverse effect on asset prices.

This approach contrasts with much of the academic literature, which anthropomorphizes 'markets,' treating them as crucial decision-making entities: Instead of considering bank demand for risk, the literature discusses 'market appetite for risk' (e.g. Musatov & Watts 2014. See also Holmstrom and Kaplan 2001). Banks are then framed as responding to competition and to market forces when originating leveraged loans (e.g. CBO 1994; Borio 1990a: 37).

When banks (or more precisely bankers) are framed as making active decisions about what loans to make, and when there is a substantial risk that if they make bad decisions and fund the wrong kinds of activities they will cause financial instability and/or asset price bubbles, then the growth of a new market, such as

<sup>&</sup>lt;sup>35</sup> The real bills doctrine was an early means of exercising such control (see Sissoko 2016; Sissoko 2022).

<sup>&</sup>lt;sup>36</sup> Sissoko (forthcoming) relates banking theory to network effects models (see e.g. Ferrell & Klemperer 2007) and to circuit theory (see e.g. Graziani 2003).

<sup>&</sup>lt;sup>37</sup> This was the phenomenon that the reals bills doctrine sought to address (see Sissoko 2016; Smith 1776: II, ii).

the leveraged loan market, needs to be evaluated with some care. There is now no presumption that 'market' phenomena will be efficiency-enhancing, and bank loans are now viewed as a proximate cause of economic distortions.

Since banks have access to cheap funding that is subsidized by the public's need for a medium of exchange, and too-big-to-fail banks have even cheaper and more abundant access to funding (Santos 2014, Gudmundsson 2016), when these banks decide to direct their vast resources into the finance of buyouts, then buyouts will take place. As central bankers have understood for centuries, while bank-driven credit flows provide essential support for economic activity, they can also create serious distortions in the economy because economic sectors will grow where credit is made available. In the case of buyouts, once people in finance, and people close enough to finance to be able to figure out how to get into it (e.g. Alexander 2018 on Vincent Naimoli), learned what types of deals the banks were interested in financing, there was no lack of 'competition' for leveraged loans. Buyout firms, and indeed different types of buyouts, proliferated precisely because the financing was there to make them happen.

In short, banking theory supports the view that the leveraged buyout boom was a bank-driven phenomenon – even while acknowledging that it could not have happened without certain antecedents such as the evolution of financial structuring activities (Knafo and Dutta 2019), the growth of junk bonds (Lazonick & O'Sullivan 1997), or the change in US law permitting pension funds to invest in buyout funds and hedge funds (Fenn et al. 1995: 10-11). Notably, this view is supported by contemporaries who also took the view that buyouts were a lender-driven phenomenon (Thackray 1986).

#### Private equity and economic distortions

Are there any economic distortions that may have developed due to the decision of the largest US banks to use the syndicated loan structure to finance buyouts? First, this distortion contributes to the increase in the flow of US national income to the financial sector, and, second, there are currently concerns that the large segment of the US corporate sector that is owned by private equity is so heavily indebted that significant defaults will be the ultimate outcome of this lending bubble.

#### The billionaire factory

The bank-driven growth of buyouts imposes heavy debt loads on the buyout targets in order to repurpose the corporation away from building long-term value and towards making payments to creditors (including the banks) and to shareholders (see section I above).

Historically the corporate form was created in order to make it easier to raise the very large sums of money needed for largescale business endeavors. It has funded essential infrastructure such as railroads

31

and trans-oceanic cables, as well large-scale manufacturing firms of every variety. By the mid-20<sup>th</sup> century corporations were integral to every part of American life.

The buyout takes the extraordinary publicly-subsidized capacity of corporations to raise large sums of money and directs the benefits of that capacity to senior creditors and a small group of insiders: this is the billionaire factory (Phalippou 2020; see also Shleifer and Summers 1988; Osterberg 1989). Thus, an important effect of the bank-driven flow of credit into leveraged loans is that a select group of financiers was enabled by the banks to arbitrage the corporate form and to use it to raise vast sums of money for the benefit of that select group of financiers, as well as the bankers. The fact that buyouts are designed to divert the benefits of the corporate structure to the banks and a small group of insiders, the private equity firms that are the general partners in the buyout funds, can surely explain at least in part why, since the 1980s, an increasing share of gross national income has flowed to the financial industry (Philippon 2015; Morris and Phalippou 2020).

### Is private equity sponsored debt a bubble?

The normalization of the buyout phenomenon has transformed the economic structure of the US economy, as a significant portion of US corporations are now heavily indebted and controlled by private equity. There are on-going concerns that the debt associated with these transactions is unsustainable, that is, that significant defaults will be the ultimate outcome of this lending bubble.

In 2019 private equity funds (including buyout funds and venture capital funds) had \$1.175 trillion in assets under management and another \$680 billion in commitments that had not been drawn. Assuming equity comprised on average 30% of the value of the companies controlled, then in 2019 private equity controlled companies were worth almost \$4 trillion.<sup>38</sup> At the end of 2019, the capitalization of US public equity markets was \$38 trillion. Thus, it is reasonable to estimate that private equity in 2019 controlled companies worth more than 10% of the US stock market. When committed capital is taken into account, this would allow private equity funds to buy another 6% of the US stock market (see Mauboussin and Callahan 2020: 5).

Furthermore, the aggregate value of private equity deals in the US increased each year from 2016 to 2019, when it reached \$755 billion. While it dropped in 2020, in 2021 the value jumped to over \$1 trillion. This amounted in a single year to 2% of US stock market capitalization. Due to rising interest rates, private

<sup>&</sup>lt;sup>38</sup> While buyouts since 2000 have used more equity than in the past, they are also much more likely to be purchasers of already leveraged companies than in the past as secondary buyouts have dramatically increased and now account for more than 50% of firm exits (see Mauboussin and Callahan 2020: 38-39).

equity activity has been dropping on a quarterly basis since the end of 2021, but was still \$975 billion in 2022 (SIFMA 2022: 10; SIFMA 2023: 13).

To understand the significance of this buyout activity, recall that private equity controlled 0% of US companies forty years ago and less than 5% twenty years ago (Mauboussin and Callahan 2020: 30). In short, by the numbers private equity has risen very quickly to control a significant portion of the US corporate landscape, and is positioned to continue displacing the public corporate form at a rapid pace.<sup>39</sup>

Needless to say, this private equity control has been accompanied by a significant increase in the debt carried by mid- and large-tier non-financial corporations (Holmstrom and Kaplan 2001; Davis 2016; Aldasoro et al. 2021). The consequences of this phenomenon are significant. In the 21<sup>st</sup> century central banks have found themselves repeatedly supporting the value of putatively 'private' credit in ways that were unthinkable a few decades ago.

After the 2007-09 financial crisis Federal Reserve policy rescued the leveraged debt markets from disaster. The initial policy of low interest rates 'for an extended period' that the Fed adopted in March 2009 was designed to address weaknesses in credit markets and to make it easier to refinance debt, both mortgages and corporate debt (FOMC 2009). What mattered most to leveraged debt markets, however, was Fed policy from 2011-14 when the maturing debt would need to be refinanced. By August 2011 the Fed had adopted a policy of holding interest rates at zero 'through mid-2013,' and an explicit purpose of this prolonged zero interest rate policy was to reassure financial markets that the policy would extend for more than three or four months (FOMC 2011: 128, 135). Then in early January 2012 the Fed extended the promise to maintain zero rates to the end of 2014 – or for three years. This extension of the zero interest rate policy was adopted explicitly to 'exert some downward pressure on market interest rates' (FOMC 2012a: 54, 167-68). Needless to say, this policy was extremely important to the ease with which both leveraged loans and junk bonds were refinanced.

The commitment to an extended period of low rates was not withdrawn until January 2015, when the statement was carefully worded to make it clear that it would be several meetings before interest rates were raised.<sup>40</sup> By this time the vast amount of high yield debt that required refinancing had already been dealt with. Only in December 2015 did the Fed finally raise interest rates.

<sup>&</sup>lt;sup>39</sup> Indeed, Mauboussin and Callahan 2020 posit that even more funds may flow into private equity in the 2020s (p. 53).

<sup>&</sup>lt;sup>40</sup> Specifically the statement read that "the Committee judges that it can be patient in beginning to normalize the stance of monetary policy" (FOMC 2015).

A case study of 'the best leveraged buyout ever' by Austin and Phalippou (2023) makes clear how the leveraged buyouts of the 2000s were heavily reliant on the Fed for their ultimate success. The buyout of Hilton Hotels in 2007 by the Blackstone Group, in which Blackstone earned a total of \$14 billion in capital gains, benefited dramatically from the decline in interest rates – which accounted for as much as 100% of the gains from the deal when evaluated in net present value terms (Austin and Phalippou 2023: Table 6). In other words, the Federal Reserve's adoption of a policy of low interest rates for a period of three years or more served in particular to support the value of highly leveraged firms with adjustable rate debt. Companies with less leverage received smaller benefits from this policy.

Furthermore, in response to the Covid crisis in 2020, the Fed provided support that was specifically directed towards the debt issues of private equity-owned companies. Leveraged loans were supported by making the senior, and largest, tranche of a CLO eligible as collateral at Fed facilities.<sup>41</sup> The junk bond market was directly supported by Fed purchases of high-yield ETFs that accounted for approximately one-eighth of all Secondary Market Corporate Credit Facility ETF purchases. In addition, even though the Primary Market Corporate Credit Facility apparently did not make any purchases, the fact that syndicated loans were eligible for purchase at the Fed almost certainly helped to support their value.

In this environment, pervasive high corporate debt is an on-going matter of concern particularly in an environment where interest rates are increasing (Smith and Cannon 2018; Platt et al. 2023). Whether or not this debt is ultimately deemed to have been a bubble may however depend on how long the central banks continue to provide support for this debt.

# V. Conclusion: Bank-driven financial innovation today

The previous section explained how bank-driven buyouts played an important role in redirecting US national income to the financial sector and may have generated a bubble in private equity sponsored debt. The example of bank-driven buyouts also provides a new framework for understanding financial innovation. When banks play a significant role in funding a financial innovation, there is a risk that bank-funding is creating a distortion in the economy. Here, we briefly discuss two candidates for further study, the collateralized loan obligation (CLO) and 'private credit.'

From the late 1990s banks began to transfer leveraged loans to non-bank investors (after collecting fees for origination). To do so, the banks relied on structured finance vehicles in the form of CLOs, as depicted

<sup>&</sup>lt;sup>41</sup> This collateral was accepted at the Primary Dealer Credit Facility and the Term Asset-Backed Securities Loan Facility, two 2007-09 crisis facilities that were re-opened in 2020.



in Chart 9 (Bullock 2008; Mauboussin and Callahan 2020: 20).



CLOs allowed the volume of leveraged loans to grow dramatically from 2003 to 2007 as financial structuring was used to increase demand for the product (Monga 2009; see also Mauboussin and Callahan 2020: 20; BIS QR Sept 2019 p. 11). All of the big names in private equity played a role in the buyout boom of 2005-07: Apollo Management LP, Bain Capital LLC, Blackstone Group LP, Kohlberg Kravis Roberts & Co, and others relied on this debt to finance huge deals that were considered 'mega-buyouts' at the time (Monga 2009; Schwartz 2010). A private equity executive's evaluation was that 'The PE firms were not investing in specific industries. They were investing in the capital markets' (Monga 2009).

Ultimately, however, leveraged loan defaults peaked in 2009 and the market did not experience a collapse at all similar to the subprime mortgage backed security or CDO markets. This can be explained by the Federal Reserve's extremely prolonged low interest rate policy, as was discussed above. This raises the question whether bank-sponsored CLOs are also playing a role in the distortions discussed above: the increase in the flow of US national income to the financial sector and the risk that private equity company debt will turn out to be a bubble. This is however a topic for future work.

A more recent financial innovation is an asset class called 'private credit' which according to Moodys (2024) is 'non-bank lending to mostly private-equity-owned, middle-market companies that aren't publicly traded or issued.' In theory, this would appear to be an alternative to bank-originated leveraged loans. In practice, banks play an important role in financing the private credit firms (Zelter 2024), and rely on private credit as they relied on junk bonds to provide the supporting debt that makes its possible for a deal to be completed (Cai and Haque 2024). Indeed, the large banks typically have private credit arms that both invest in this debt and provided financing for the funds that invest in this debt (Whittall 2019; Whittall 2022). Whether private credit is another bank-driven phenomenon requires further research, but this paper indicates that it is an important question that should be investigated.

### References

Viral V. Acharya, Oliver F. Gottschalg, Moritz Hahn and Conor Kehoe, 2013. Corporate Governance and Value Creation: Evidence from Private Equity The Review of Financial Studies, February 2013, Vol. 26, No. 2 (February 2013), pp. 368-402.

Tobias Adrian and Hyun Song Shin 2010. "Liquidity and Leverage", Journal of Financial Intermediation, 19, pp. 418–37.

James F. Albertus and Matthew Denes 2023. Private Equity Fund Debt: Agency Costs and Cash Flow Management. SSRN working paper.

Iñaki Aldasoro, Bryan Hardy & Nikola Tarashev 2021. Corporate debt: post-GFC through the pandemic. BIS Quarterly Review, June 2021.

Brian Alexander. 2018. Glass House: The 1% Economy and the Shattering of the All-American Town. Macmillan

T.E. Allen 1971. Note for record: Swiss Official Attitude to International Lending in Swiss Francs. July 7, 1971. BoE Archive 8A406/3

T.J. Allen (1990). Developments in the international syndicated loan market in the 1980s. Bank of England Quarterly Review, February, pp. 71-77.

Yener Altunbas, Blaise Gadanecz & Alper Kara 2006. The Evolution of Syndicated Loan Markets. The Service Industries Journal, Vol.26, No.6, September 2006, pp.689–707.

Appelbaum, Eileen, and Rosemary Batt. 2014.Private Equity at Work : When Wall Street Manages Main Street, Russell Sage Foundation.

Neroli Austin & Ludovic Phalippou 2023. Decomposing value gains – The case of the best leveraged buy-out ever. Journal of Corporate Finance. 81 https://doi.org/10.1016/j.jcorpfin.2022.102317.

Brian Ayash and Mahdi Rastad, "Leveraged Buyouts and Financial Distress" Finance Research Letters 38 (2021) 1014-52. https://doi.org/10.1016/j.frl.2020.101452

Kenneth M. Ayotte, Edward R. Morrison 2009. Creditor Control and Conflict in Chapter 11. Journal of Legal Analysis, Volume 1, Issue 2, Summer 2009, Pages 511–551, https://doi.org/10.1093/jla/1.2.511

BIS 2019. Structured finance then and now: a comparison of CDOs and CLOs. Quarterly Review. September 2019.

Bank of England 1973. Report: The International Capital Markets – 1972. 26 January 1974. BoE Archive 8A406/4.

Bank of England 1973b. Summary Restricted to participants: consultation with private-bankers on eurocurrency lending to developing countries. BoE archive 8A406/5

Bank of England 1974. Report: The International Capital Markets – 1973. 25 January 1974. BoE Archive 8A406/5.

Bank of England 1978. Confidential Report: The SDR as a possible denominator of international debts. May 24 1978. BoE Archive 4A39/30.

Ulrich Bindseil (2019). Central Banking before 1800: A Rehabilitation. Oxford University Press.

I D Bond (1985). The syndicated credits market. Bank of England Discussion paper No 22. March.

Richard Booth (1991). FEDERALISM AND THE MARKET FOR CORPORATE CONTROL Washington U Law Qly 69, pp 411-43.

Claudio Borio 1990a. LEVERAGE AND FINANCING OF NON-FINANCIAL COMPANIES: AN INTERNATIONAL PERSPECTIVE. BIS Economic Papers No 27.

Claudio Borio 1990b. BANKS' INVOLVEMENT IN HIGHLY LEVERAGED TRANSACTIONS. BIS Economic Papers No 28

Boughton, James (2001). The Silent Revolution: The International Monetary Fund 1979-1989. IMF, Washington DC

Brunnermeier, M. & Pedersen, L. (2009). Market liquidity and funding liquidity, Review of Financial Studies, 22(6), 2201-2238.

Nicole Bullock (2008). Once staid US loan market braced for further hardship. Financial Times. Nov 4 2008. http://www.ft.com/cms/s/0/72cb2c06-aa10-11dd-958b-000077b07658.html

Burns, Arthur (1974). Speech: Maintaining the Soundness of our Banking System. October 21, 1974.

Burns, Arthur (1977). Speech: The need for order in international finance. April 12, 1977.

Paul Butler 2001. The Alchemy of LBOs. McKinsey Quarterly, No. 2, pp 140-51

Cai, Fang, and Sharjil Haque (2024). "Private Credit: Characteristics and Risks," FEDS Notes. Washington: Board of Governors of the Federal Reserve System, February 23, 2024, https://doi.org/10.17016/2380-7172.3462.

Nicola Cetorelli, Benjamin H. Mandel, and Lindsay Mollineaux 2012. The evolution of banks and financial intermediation: Framing the analysis. Federal Reserve Bank of New York Economic Policy Review, 18(2), 1-12.

Brett Christophers (2023). Our lives in their portfolios: Why asset managers own the world. Verso.

Philip Coldwell 1976. Protecting the public interest in banking institutions. Speech at National Economists Club Seminar, Washington DC, May 19 1976.

Congressional Budget Office 1994. THE CHANGING BUSINESS OF BANKING: A STUDY OF FAILED BANKS FROM 1987 TO 1992. June 1994.

Leila E. Davis (2016) Identifying the "financialization" of the nonfinancial corporation in the U.S. economy: A decomposition of firm-level balance sheets, Journal of Post Keynesian Economics, 39:1, 115-141, DOI: 10.1080/01603477.2015.1116370

Robert DeYoung and Tara Rice 2004. How do banks make money? The fallacies of fee income. FRB Chicago Economic Perspectives, Q4, pp 34-51.

Dodd, E. M. Jr (1932). For whom are corporate managers trustees? Harvard Law Review, 45, 1145-1163.

Eleanor Duncan 2018. Banks can 'do what they want' in leveraged lending: Otting. Feb 28 2018. https://in.mobile.reuters.com/article/amp/idINKCN1GC0B5

Economist 2008. Barbarians at the gate 2.0. Oct 28 2008. https://www.economist.com/business/2008/10/28/barbarians-at-the-gate-20

Edgley 1973. Memo titled: OECD Commmittee on Financial Markets Developing Countries and Eurocredits. Oct 26 1973. BoE Archive 8A406/5.

Farrell, J, Klemperer, P. (2007): Coordination and Lock-In: Competition with Switching Costs and Network Effects, in: Armstrong, M., Porter, R. (eds.), Handbook of Industrial Organization, vol. 3, 967-2072.

Fed Board 1992. Interagency notice on Highly Leveraged Transactions. February 6. Effective Date February 12.

FDIC (1997). History of the Eighties – Lessons for the Future.

FOMC 2009. Meeting transcript. March 17-18 2009.

FOMC 2011. Meeting transcript. August 9 2011.

FOMC 2012. Meeting transcript. January 24-25 2012.

George Fenn, Nellie Liang, and Stephen Prowse 1995. The Economics of the Private Equity Market on early growth of PE. Federal Reserve. 1995.

Freeman RE (1984) Strategic Management: A Stakeholder Approach. Boston, MA: Pitman.

Brett D. Fromson and Jerry Knight May 16, 1993 The Saving of Citibank. Washington Post. https://www.washingtonpost.com/archive/politics/1993/05/16/the-saving-of-citibank/64323ac1-db21-43e4-b89e-a3171324c202/ viewed 6 July 2023.

Rodney Galpin 1973. Memo to Mr. Edgley titled Medium Term Euro-Currency Credits. Oct 29 1973. BoE Archive 8A406/5.

Michelle Garfinkel 1989. Causes and Consequences of Leveraged Buyouts. FRB St Louis Review, Sept/Oct, pp. 23-34

Gerschenkron, A. (1962), Economic Backwardness in Historical Perspective, Harvard University Press, Cambridge, MA.

Douglas H. Ginsburg and John F. Robinson, 1986 The Case against Federal Intervention in the Market for Corporate Control. The Brookings Review, Winter - Spring, 1986, Vol. 4, No. 2 (Winter - Spring, 1986), pp. 9-14.

Graziani, A. (2003): The Monetary Theory of Production. Cambridge: Cambridge University Press

Tryggvi Gudmundsson 2016. Whose Credit Line is it Anyway: An Update on Banks' Implicit Subsidies. IMF Working Paper WP/16/224

Hayes, Thomas (1983). The two sides of Security Pacific. New York Times, July 17, https://www.nytimes.com/1983/07/17/business/the-two-sides-of-security-pacific.html

Bengt Holmstrom and Steven N. Kaplan 2001. Corporate Governance and Merger Activity in the United States: Making Sense of the 1980s and 1990s, Journal of Economic Perspectives—Volume 15, Number 2—Spring 2001—Pages 121–144.

Karen Horn 1984. Comments at Meeting of the Federal Open Market Committee May 21-22, 1984.

International Finance Corporation 1973. Annual Report. Extract available in BoE Archive 8A406/5.

Jarrell, Gregg A., James A. Brickley, and Jeifry M. Netter. 1988 "The Market for Corporate Control: The Empirical Evidence Since 1980:' Journal of Economic Perspectives (Winter 1988), pp. 49-68.

Jensen, Michael C. 1986 "Agency Costs of Free Cash Flow, Corporate Finance, and Takeovert" American Economic Review (May 1986), pp. 323-29

Jensen, Michael C. 1988 'Takeovers: Their Causes and Consequences:' Journal of Economic Perspectives (Winter 1988), pp. 21-48.

Jensen, Michael C., and William H. Meckling. 1976 "Theory of the Firm: Managerial Behavior, Agency Costs and Ownership Structure' Journal of Financial Economics (October 1976), pp. 305-60.

Jensen, Michael C., and Richard S. Ruback. 1983 "The Market for Corporate Control: The Scientific Evidence, Journal of Financial Economics (April 1983), pp. 5-50.

Kaplan & Stromberg 2009 Leveraged Buyouts and Private Equity JEP 23(1) pp. 121-46.

Ethan Kapstein (1994). Governing the global economy: international finance and the state. Harvard U Press: Cambridge.

Knafo, S., & Dutta, S. J. (2016). Patient capital and the financialization of management. Socio- Economic Review, 14 (4), 771–788. doi:10.1093/ser/mww027

Samuel Knafo & Sahil Jai Dutta (2019): The myth of the shareholder revolution and the financialization of the firm, Review of International Political Economy, DOI: 10.1080/09692290.2019.1649293

Jerry Knight (1988). REGULATORS WORRY ABOUT RISK IN FINANCING OF BIG TAKEOVERS. Washington Post. Nov 28 1988.

Lazonick, W. and O'Sullivan, M. (1997) 'Finance and Industrial Development. Part I: The United States and the United Kingdom', Financial History Review, 4, 7–29.

Peter Lee 1992. Is Citibank back from the dead? Euromoney. Dec 1 1992. https://www.euromoney.com/article/b1fhmp5p2111kg/is-citi-back-from-the-dead viewed 6 July 2023.

Kenneth Lehn and Annette Poulsen 1989. Free Cash Flow and Stockholder Gains in Going Private Transactions, The Journal of Finance, Jul., 1989, Vol. 44, No. 3, Papers and Proceedings of the Forty-Eighth Annual Meeting of the American Finance Association, New York, New York, December 28-30, 1988 (Jul., 1989), pp. 771-787.

Los Angeles Times (1988). Big Banks Speak Out in Support of LBOs. December 9.

Stephen Lubben 2015. The Board's Duty to Keep Its Options Open. University of Illinois Law Review 2015(2). Pp 817-30.

Michael J. Mauboussin and Dan Callahan (2020). Public to Private Equity in the United States: A Long-Term Look. Morgan Stanley. August 4, 2020.

Vipal Monga (2009). That worrying wall of debt. The Deal. June 5, 2009. http://www.thedeal.com/newsweekly/features/that-worrying-wall-of-debt.php

Monnet, Eric (2018). Controlling Credit: Central Banking and the Planned Economy in Postwar France, 1948–1973. Cambridge University Press: New York.

Moodys (2024). What is private credit? (website) available at <u>https://www.moodys.com/web/en/us/private-credit.html</u> viewed February 26 2024.

Peter Morris and Ludovic Phalippou 2020. Thirty years after Jensen's prediction: Is private equity a superior form of ownership? Oxford Review of Economic Policy, Volume 36, Number 2, 2020, pp. 291–313

Peter Morris, Ludovic Phalippou, and Betty Wu 2022. How Deadly Is Financial Leverage? Evidence from Care Homes during the COVID-19 crisis. SSRN working paper.

Alex Musatov and William Watts 2014. Despite Cautionary Guidance, Leveraged Loans Reach New Highs. Dallas Fed Economic Letter. VOL. 9, NO. 10. SEPTEMBER 2014.

William Osterberg 1989 LBOs and Conflicts of Interest, Economic Commentary Federal Reserve Bank of Cleveland, August 15.

William Osterberg 1993. Bank Exposure to Highly Leveraged Transactions, Economic Commentary Federal Reserve Bank of Cleveland, January 15.

George Parker, James Pickard, and Gill Plimmer, 29 June 2023, "Water Industry plunges into deepest crisis" FT

Phalippou, Ludovic. 2020. An Inconvenient Fact: Private Equity Returns & The Billionaire Factory (June 10, 2020). University of Oxford, Said Business School, Working Paper, Available at SSRN: https://ssrn.com/abstract=3623820 or http://dx.doi.org/10.2139/ssrn.3623820

Ludovic Phalippou, Christian Rauch, Marc Umber (2018). Private equity portfolio company fees. Journal of Financial Economics 129 (2018) 559–585.

Thomas Philippon (2015). Has the US Finance Industry Become Less Efficient? On the Theory and Measurement of Financial Intermediation. American Economic Review 2015, 105(4): 1408-1438. http://dx.doi.org/10.1257/aer.20120578

Eric Platt, Antoine Gara and Harriet Clarfelt 2023. Moody's warns of 'serious challenge' to \$1.4tn private credit market. FT. 23 June 2023. https://www.ft.com/content/de6eb245-80ff-4f9c-b2ed-7a415538512b

Gill Plimmer and Nic Fildes 28 June 2023 "Managed by Macquarie" FT; Bryce Elder 28 June 2023 "A quick guide to Thames Water's serpentine capital structure" FT https://www.ft.com/content/ee57540b-072c-488e-9eae-8cc11886e2c4

Michael Robinson, 5 Sept 2017, "How Macquarie bank left Thames Water with extra £2bn debt" BBC News, https://www.bbc.co.uk/news/business-41152516

N.M. Rothschilds & Sons Ltd 1971. The 'Cecu' unit. Sept 16 1971. In BoE Archives 8A406/3.

João Santos 2014. Evidence from the Bond Market on Banks' "Too-Big-to-Fail" Subsidy. FRBNY Economic Policy Review. December 2014 Volume 20, Number 2

Schumpeter, J. A. (1939), Business Cycles: A Theoretical Statistical and Historical Analysis of the Capitalist Process (2 vols), New York: McGraw-Hill.

Nelson Schwartz 2010. Corporate Debt Coming Due May Squeeze Credit. NYT March 15 2010 http://www.nytimes.com/2010/03/16/business/16debt.html

L. William Seidman 1989. "The Impact of Highly Leveraged Transaction upon the Banking Industry" Testimony before the House Committee on Ways and Means, January 31 1989.

Senate Banking Act Report (1933), Senate Report No. 73-77.

John Shad 1984. The Leveraging of America. Securities and Exchange Commission. Speech. NYC. June 7, 1984. https://www.sechistorical.org/collection/papers/1980/1984\_0607\_LeveragingShad.pdf

Shleifer, Andrei, and Lawrence H Summers. 1988. "Breach of Trust in Hostile Takeovers." Corporate Takeovers: Causes and Consequences., edited by Alan J Auerbach, 33-56. Chicago: University of Chicago Press.

Andrei Shleifer and Lawrence H. Summers 1990. The Noise Trader Approach to Finance, Journal of Economic Perspectives- Volume 4, Number 2- Spring 1990- Pages 19-33.

Andrei Shleifer and Robert Vishny 1990. The Takeover Wave of the 1980s. Science. 17 August. Pp 745-49.

SIFMA 2022. SIFMA Research Quarterly – 4Q21: Primary Market: US Equity Capital Formation, Secondary Markets: US Cash Equities, ETFs, and Multi-Listed Options. January 2022.

SIFMA 2023. Quarterly Report: US Equity & Related, 1Q23. April 2023.

Sissoko, C. 2016. How to stabilize the banking system: Lessons from the pre-1914 London money market. Financial History Review 23(1), 1 - 20.

Sissoko, C. 2017. The plight of modern markets: How universal banking undermines capital markets, Economic Notes, 46(1), pp. 53–104.

Sissoko, Carolyn 2021. "The Nature of Money in a Convertible Currency World." Review of Economic Analysis, 13(1), pp. 1-43.

Sissoko, C 2022. "Becoming a central bank: The development of the Bank of England's private sector lending policies during the Restriction" Economic History Review 75(2) pp 601-32. https://doi.org/10.1111/ehr.13110

Sissoko, C. forthcoming. The Bank-Centered View of the Financial System: Why Banks Are Different. European Journal of Economics and Economic Policies.

Adam Smith 1776. The Wealth of Nations.

Molly Smith and Christopher Cannon 2019. A \$1 Trillion Powder Keg Threatens the Corporate Bond Market. October 11, 2018. https://www.bloomberg.com/graphics/2018-almost-junk-credit-ratings/

Joan Spero 1980. The Failure of the Franklin National Bank: Challenge to the International Banking System. Columbia University Press.

Irvine H. Sprague 1986. Bailout. Basic Books.

John Thackray 1986. Leveraged buyouts: The LBO craze flourishes amid warnings of disaster. Euromoney. Jan 2, 1986. https://www.euromoney.com/article/b13206c1922hwt/leveraged-buyouts-the-lbo-craze-flourishes-amid-warnings-of-disaster

Juan Carlos Torre 2021. Diario de una temporada en el quinto piso: Episodios de política económica en los años de Alfonsín. EDHASA, Oct 14, 2021

Stefano Ugolini (2017). The Evolution of Central Banking: Theory and History. Palgrave Macmillan.

Paul Volcker (1980). Speech: The Recycling Problem Revisited. March 1, 1980.

Henry Wallich (1981). Speech. LDC debt – to worry or not to worry. June 2 1981.

Whittall, Christopher 2019. Deutsche Bank's bold gambit: betting big on private credit. International Financing Review, no. 2302, 30 September 2019. <u>https://www.ifre.com/story/2077623/deutsche-banks-bold-gambit-betting-big-on-private-credit-15n26e3k3</u>

Whittall, Christopher 2022. If you can't beat them, join them: banks cash in on private credit boom. International Financing Review, no. 2447, 19 Aug 2022. <u>https://www.ifre.com/story/3480927/if-you-cant-beat-them-join-them-banks-cash-in-on-private-credit-boom-59f3ckl7rz</u>

Wienke, Robert O., 1994, Loan Syndications and Participations: Trends and Tactics, Commercial Lending Review, Vol 9, Issue 2: 4-10.

Arthur E. Wilmarth, Jr., 2002 The Transformation of the U.S. Financial Services Industry, 1975-2000: Competition, Consolidation, and Increased Risks, 2002 U. ILL. L. REV. 215 –

Arthur E. Wilmarth, Jr. 2014. Citigroup: A Case Study in Managerial and Regulatory Failures, 47 IND. L. REV. pp. 69-137.

J. K. Wright 1973. Letter from Overseas Development Administration to S.W. Payton. 9 August 1973. BoE Archive 8A406/5.

Zelter, Jim (2024). Apollo's Zelter sees pockets of stress in the economy. Bloomberg video. February 13 2024. <u>https://www.bloomberg.com/news/videos/2024-02-13/apollo-s-zelter-sees-pockets-of-stress-in-economy</u>